Title: A mouse model of sitosterolemia; absence of Abcg8/sterolin-2 results in failure to secrete biliary cholesterol

Authors:

Dr Eric L Klett (klettel@musc.edu)
Kangmo Lu (Kangmo-Lu@ouhsc.edu)
Astrid Kosters (A.Kosters@amc.uva.nl)
Edwin Vink (E.Vink@amc.uva.nl)
Mi-Hye Lee (leemih@musc.edu)
Michael Altenburg (mikelly@email.unc.edu)
Sarah Shefer (salenge@umdnj.edu)
Ashok K Batta (ashkbatta@hotmail.com)
Dr Hongwei Yu (yuh@musc.edu)
Mr Jianliang Chen (chenjl@musc.edu)
Richard Klein (kleinrl@musc.edu)
Norbert Looije (N.Looije@amc.uva.nl)
Ronald Oude-Elferink (r.p.oude-elferink@amc.uva.nl)
Albert K Groen (a.k.groen@amc.uva.nl)
Nobuyo Maeda (nobuyo.maeda@pathology.unc.edu)
Gerald Salen (salenge@umdnj.edu)
Dr Shailendra B Patel (patelsb@musc.edu)

Version: 4 Date: 27 Feb 2004

We would like to resubmit our manuscript entitled 'A mouse model of sitosterolemia; absence of Abcg8 results in failure to secrete biliary cholesterol' for publication in BMC Medicine. The formatting changes requested have been performed. These include: 1) Linking each author to their corresponding affiliation with the use of superscripts, 2) Countries have been added to all affiliation addresses, 3) The corresponding author is now only Eric L. Klett, 4) Email addresses for all authors is now included on the title page, 5) Reference citations now include all authors listed and referencing is per BMC End Note formatting, 6) Table 1 has been changed to include a border in all cells, 7) All figures have been cropped to minimize white space around the image.